Author's response to reviews

Title: Impact of adjuvant chemotherapy for gliomatosis cerebri

Authors:

Doo-Sik Kong (kds026@skku.edu)
Sung Tae Kim (stkim08@skku.edu)
Jung-II Lee (jilee@skku.edu)
Yeon-Lim Suh (yl.suh@samsung.com)
Do Hoon Lim (dh8lim@skku.edu)
Won Seog Kim (wskimsmc@skku.edu)
Ki-Hoon Kwon (kkw71@skku.edu)
Kwan Park (kwanpark@skku.edu)
Jong Hyun Kim (jionghkim@skku.edu)
Do-Hyun Nam (nsnam@skku.edu)

Version: 6 Date: 3 August 2010

Author's response to reviews: see over
Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

: We re-write our authors’ affiliation according to the journal’s principles

1 Department of Neurosurgery, Samsung Medical Center, Sungkyunkwan University School of Medicine, Seoul, Korea
2 Department of Radiology, Center for Imaging Science, Samsung Medical Center, Sungkyunkwan University School of Medicine, Seoul, Korea
3 Department of Pathology, Samsung Medical Center, Sungkyunkwan University School of Medicine, Seoul, Korea
4 Department of Radiation Oncology, Samsung Medical Center, Sungkyunkwan University School of Medicine, Seoul, Korea
5 Division of Hematology-Oncology, Department of Medicine, Samsung Medical Center, Sungkyunkwan University School of Medicine, Seoul, Korea

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Qualifications: Please remove the authors’ qualifications (e.g. PhD / MD) or job titles from the manuscript and submissions system.

: Doo-Sik Kong 1*, Sung Tae Kim 2*, Jung-Il Lee 1, Yeon-Lim Suh 3, Do Hoon Lim 4, Won Seog Kim 5, Ki-Hoon Kwon 1, Kwan Park 1, Jong Hyun Kim 1, Do-Hyun Nam 1

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efg@generic.co.uk
IJ: IJKL@corporation.com

DK: kds026@skku.edu
SK: stkim08@skku.edu
JL: jilee@skku.edu
YS: ylsuh76@skku.edu
DL: dh8lim@skku.edu
WK: wskimsmc@skku.edu
KK: kihoonkwon71@skku.edu
KP: kwanpark@skku.edu
JK: jonghkim@skku.edu
DN: dh8lim@skku.edu

Funding: Please move the statement of funding to the Acknowledgements section and remove it from the title page.

: We rearrange the acknowledgement

Title page: Please remove the running title and keywords from the title page.

We remove the running title, and keywords
Abstract: please make sure that the abstract section has the heading 'Abstract'.

: Abstract

Background. Gliomatosis cerebri (GC) is characterized by a diffuse infiltration of tumor cells throughout CNS, however, few details are available about the chemotherapeutic effect on GC. The aim of this study was to investigate its clinical course and to determine the efficacy of chemotherapy for GC.

Abstract: Please remove the keywords from the abstract.
: We removed the keywords

Abstract: please rename the section 'Objective' in the abstract 'Background'.
: we changed into the background

Introduction: Please rename this section 'Background'.
: Background

Gliomatosis cerebri (GC) is a rare variant of glioma. It is characterized by a diffuse infiltration of malignant glial cells throughout large regions of the central nervous system [1-6] and a relative preservation of the neural architecture. Since Nevin first used the term gliomatosis cerebri in 1938, describing the histological finding of glial overgrowth

Clinical Materials and methods: Please rename this section 'Methods'.
: Methods

Including criteria
We retrospectively identified 45 consecutive patients diagnosed with GC who were examined at a single institution between Jan. 1999 and Dec. 2004. Of these, 37 patients were

Structure: Please insert the competing interests before the authors' contributions and the acknowledgements after the authors' contributions.
:We rearranged it.

: Competing interests

The authors declare that they have no competing interests.

Authors’ contributions
DN was responsible for the conception and design of the study; DK and SK performed the collection and assembly of data; DK, SK, JL, KK, KP, and JK performed the statistical analysis and interpretation of data; DK, SK and YS participated in writing the manuscript; DL, WK, and DN revised the manuscript. All authors read and approved the final manuscript.

Acknowledgements
This study was supported by a grant from Samsung Biomedical Research Institute (C-A7-410-1), by a grant of the Korea Healthcare technology R&D Project, Ministry for Health, Welfare & Family Affairs, Republic of Korea (A092255) and the Korea Science and Engineering foundation (KOSEF) grant funded by Korea government (MOST) (No. M1064100104-06N4100-10410).

References: the reference list should contain up to 30 authors' names for each citation. The term 'et al' should not be used if there are fewer than 30 authors.
References: please provide the full citation details for references 14 and 15. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references.

Additional files: Please remove the additional file from the manuscript.
we removed it

Figures: Please ensure all mention of figures is consistent (it should be Figure or Fig but not a mixture of both).
: we confirmed it

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.